\documentclass[11pt]{article}
 \setlength{\textheight}{10.4in}
 \setlength{\topmargin}{-1in}
 \setlength{\headheight}{0in}
 \setlength{\topsep}{0in}
 \setlength{\marginparwidth}{0in}
 \setlength{\marginparsep}{0in}
% \setlength{\footskip}{4pc}
%\setlength{\footheight}{0in}
 \setlength{\oddsidemargin}{0in}
 \setlength{\evensidemargin}{0in}
 \setlength{\textwidth}{6.4in}
 \setlength{\parskip}{4pt}
 \setlength{\parindent}{0pt}
\parindent=0in
\parskip=1pc
\begin{document}
\thispagestyle{empty}
\section*{Quizzes}
A Quiz is a take-home problem set, one per week. They are
graded like exams, score 0-100 on each problem. 

Quizzes are assigned weekly, and due the next week. If it is week 5 and
this message appears instead of Quiz5, then there is a mistake, or else nothing is due.
If in doubt, call or send email.



\end{document}
